Author's response to reviews

Title: Patient acceptance and perceived utility of pre-consultation prevention summaries and reminders in general practice: pilot study

Authors:

Oliver R Frank (oliver.frank@adelaide.edu.au)
Nigel P Stocks (nigel.stocks@adelaide.edu.au)
Paul Aylward (paul.aylward@adelaide.edu.au)

Version: 9 Date: 3 May 2011

Author's response to reviews:

The caption for Figure 1 does already say that the patient's name is fictitious.

We have changed the patient's name to Fictitious Patient to make this more clear, and have attached the revised Figure 1.

Thanks for your help.